Author's response to reviews

Title: Efficacy and safety of pamidronate in Modic type 1 changes: study protocol for a prospective, randomized, controlled clinical trial

Authors:

Stella Cecchetti (stellacecchetti@hotmail.fr)
Bruno Pereira (bpereira@chu-clermontferrand.fr)
Antoine Roche (aroche@chu-clermontferrand.fr)
Christophe Deschaumes (cdeschaumes@chu-clermontferrand.fr)
Dihya Abdi (dabdi@chu-clermontferrand.fr)
Emmanuel Coudeyre (ecoudeyre@chu-clermontferrand.fr)
Jean-Jacques Dubost (jjdubost@chu-clermontferrand.fr)
Sylvain Mathieu (smathieu@chu-clermontferrand.fr)
Sandrine Malochet-Guinamand (smalochet@chu-clermontferrand.fr)
Anne Tournadre (atournadre@chu-clermontferrand.fr)
Marion Couderc (mcouderc@chu-clermontferrand.fr)
Marielle Vayssade (mvayssade@chu-clermontferrand.fr)
Coline Daron (cdaron@chu-clermontferrand.fr)
Martin Soubrier (msoubrier@chu-clermontferrand.fr)

Version: 6 Date: 12 March 2014

Author's response to reviews: see over
Efficacy and safety of pamidronate in Modic type 1 changes: study protocol for a prospective, randomized, controlled clinical trial

Please find below a point-by-point response to the concerns raised following the review of the original manuscript.

Editorial’s request: As requested, we have modified Authors’ Contributions section to demonstrate that each author meets all the criteria to qualify for authorship, including point 2 of the ICMJE authorship requirements.

The authors declare that they have no conflicts of interest and have given their written permission to publish this work.

The contents of the manuscript have not been published elsewhere. Correspondence should be addressed to:

Dr Martin SOUBRIER, Service de Rhumatologie
CHU Gabriel Montpied
58 rue Montalembert
63003 CLERMONT FERRAND CEDEX 1
FRANCE

We hope that this manuscript will meet your requirements.

Sincerely

S Cecchetti, B Pereira, A Roche, C Deschaumes, D Abdi, E Coudeyre, JJ Dubost, S Mathieu, S Malochet-Guinamand, A Tournadre, M Couderc, M Vayssade, C Daron, M Soubrier